Author's response to reviews

Title: Meta-analysis of the risk of cataract in type 2 diabetes

Authors:

Li Li (lliilili2@hotmail.com)
Xiu-hua Wan (Xihuawanxw@163.com)
Guo-hong Zhao (guohongzhaog@163.com)

Version: 2 Date: 15 May 2014

Author's response to reviews: see over
Dear editor,

We would like submit our new original article entitled “Meta-analysis of the risk of cataract in type 2 diabetes” to the journal “BMC Ophthalmology”.

We announced several things as follows:

1. This is an original article, and that it has not been submitted simultaneously for publication elsewhere.

2. That the corresponding author and all of the authors have read and approved the final submitted manuscript.

3. That no portion of the manuscript, other than the abstract, has been previously published or posted in the Internet.

*Corresponding author: Li Li
Email: liliilili2@hotmail.com
Address: Department of Ophthalmology, Beijing Children’s Hospital, Capital Medical University, Beijing, 100036, China
Tel and Fax: +86-010-59716951

Thank you for any consideration of the publication of the paper!
Yours Sincerely,

Li Li